Author's response to reviews

Title: Knowledge and Attitude of Healthcare Workers about Middle East Respiratory Syndrome in multispecialty hospitals of Qassim, Saudi Arabia

Authors:

Muhammad U Khan (umair@ucsiuniversity.edu.my)
Shahjahan Shah (shahjahanshah202@gmail.com)
Akram Ahmad (akrampharma67@gmail.com)
Omotayo Fatokun (fatokun@ucsiuniversity.edu.my)

Version: 3 Date: 22 July 2014

Author's response to reviews:

The Editorial Office
BMC Public Health
BioMed Central

Dear Sir/Madam,

Thank you very much for considering our manuscript (Manuscript ID 1490086041134731). Authors have revised the manuscript in view of the changes suggested by the editorial office. Below are the description of changes made by the authors.

Title page: Added to manuscript
Ethical Statement: A subsection has been added in methodology section
Competing interest: Added to manuscript
Authors' Contribution: Section has been added to manuscript
Acknowledgement: Added as suggested
Line Numbering: Included in the text